Author's response to reviews

Title: Dietary analysis and patterns of nutritional supplement use in normal and age-related macular disease affected subjects: a prospective cross-sectional study.

Authors:

Hannah Bartlett (bartlehe@aston.ac.uk)
Frank Eperjesi (F.Eperjesi@aston.ac.uk)

Version: 2 Date: 17 August 2004

Author's response to reviews:

As no reviewer comments were received, the manuscript has been reformatted according to the pre-acceptance checklist. In particular, the layout of the references has been updated, and full journal names replaced with abbreviated versions. Some grammatical and spelling errors have been corrected.